Author's response to reviews

Title: Subjective and objective outcome in congenital clubfoot; a comparative study of 204 children

Authors:

David J Chesney (dichesney@rcsed.ac.uk)
Simon Barker (slbarker@rcsed.ac.uk)
Nicola Maffulli (n.maffulli@keele.ac.uk)

Version: 4 Date: 20 June 2007

Author's response to reviews:

further to your recent e-mail, I have made the small changes to the formating of the manuscript as requested. the authors have been linked to institutions using superscript numbers, capitalisation of headings has been changed. highlighting within teh text has been removed. the declaration that all authors have read and approved the manuscript has been added.

David Chesney